Author's response to reviews

Title: Early intervention in panic: randomized controlled trial and cost-effectiveness analysis

Authors:

Peter Meulenbeek (p.meulenbeek@compaqnet.nl)
Godelief Willemse (gwillemse@telfort.nl)
Filip Smit (fsm@trimbos.nl)
Anton van Balkom (vanbalkom@ggzba.nl)
Philip Spinhoven (SPINHOVEN@fsw.leidenuniv.nl)
Pim Cuijpers (p.cuijpers@psy.vu.nl)

Version: 3 Date: 27 November 2008

Author's response to reviews:

The Editor
BMC Trials

Date: 2008-11-27 Contact: P. Meulenbeek

Dear Editor,

Thank you very much for acceptance, in principle, of the manuscript for publication in Trials. After formatting of the manuscript by The BioMed Central Editorial Production Team, we checked the manuscript and made some very small changes.

We kindly request you to direct any correspondence on this paper to:
Peter Meulenbeek
VU-University
Hof van Parijs 45
7007 JD Doetinchem
The Netherlands
Tel. + 31 314 378 457
Email: p.meulenbeek@compaqnet.nl

We thank you for your attention.

Yours sincerely,

Peter Meulenbeek
Godelief Willemse
Filip Smit
Anton van Balkom
Philip Spinhoven
Pim Cuijpers